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SPECIAL THICKENERS, 105.t.

o, Walker P o
woter from blost-furace gas » Walker Procass, take up to 16,000 gpm of effiuent

scrubboers, loaded with washed-out flue-dust solidy
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Water treatment of
blast-furnace-gas
.serubber output can
stop stream pollution
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GAS-SCRUBAER wath water enters the thicke HireR CAKE, at or

tion well; clorified woter. then leaves ot

SLUDGE-RAKING ARMS
the scrubbar efffuent

ners, above, through o center distriby.
surfoce by woy of tha two ring  weirs

at bottom of thickeners scrape the settted-out solids from
o the thickenar conter for thélr final removal by pumps

A
>XDravo Corp in cooperation with Jones
& Laughlin's engineering dept, has two
Walker Process Clarifiow thickeners,
photos, left, p 96, sludge-pumping fa-
cilities, vacuum filier plant, and dry-

ke-handling facilities. The plant util-
l;u.h the existing gravity-flow system
vdrom the washers at the blast furnace,

duced at a rate of approximately 65 tan
in eight hours (wet basis), goes to ¢ '
sintering plant with a rated capacity of 3
100 10ns per eight hours. *
The original dust-collection system
consisted of gravity flow of wash water >
{from the blast-furnace arca to o large
U-ype settling lagoon where a lerge
per ge of the solids settled out
before the waier overflowed into the
plant sewer. This method was expen §
slve in that it required continual cla |
shelling of the muck from the lagoon -
-and loading it into railroad cars for the :
trip to the sintering plani, where jt wa .
dumped wet with other sintering:plant
ora. co-
The

water from the washer stage. :
Heart of the plant, two 105-N.dia
Walker Process thickeners, handles o
6,000.gpm Aow. One important feature
‘. is the intake arrangement: Influent en-
chn an annular ring around the large
v center well nnd then flows to the center
well through maltiple tangential diffu-

ly new pinnt, designed by

powit %
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I "5VEAMEAD receiver tonk gets sludge output
“yna-"edl it into ona of tha two efecter tanks shown above
-

ound 22.5 to 25% moisture content, travels
by conveyor to hopper. in the B0O-ton-per-hr sintering plant

from pumps,

sion perts thal produce a pronounced
rotstional movement. This diffusion
energy couses s circulor motion in the
{arge center well, which displaces down-
ward and entcrs the tonk's floor reglon.
Such s rodially Aowing current, with
equal-acting vectors ol flow displace.
ment, gives balance and lateral disiribu-
tion in the tank. Further, it serves to
concentrate the solids within the liquid.
This denser mass, legving the bottom of
the eenter well with » circulnr motion,
fans out olong the tank floor.

1n turn, clarified water—that is, water
froe of settleable solids—displaced
ahove this sirong, moving current, rises
upward and flows over into a multiple,
balanced overflow weir systam. This
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DISK-TYPE filters receive collectod' slurry from ejector tanks
at about 50% solids concenpiation ond filier it 10 0 coke form

VACUUM PUMPS produce tha roquired vacuum lor the disk-typo
fliters shown above; some 1690 clm eoch ot 20 in. vocuum

‘system. Velocity is so low and uniform,
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weir system runs uniformly around the
entire surface area of the clarification
zone. Velocity of appreach 10 the weirs,
though, is low because of the masa of
water displaced. The low-velocily level
prevents reaching back and drowing up
previously settled solids into the weir,

in fact, that previously settled precipi-,
tate particles, introduced near the over-
flow weir, settle back countercurrent to

the flow and do not overflow inlo the * ..

weirs to cause any clogging. .
Recirculation of previously preeipi-

iated solids, by withdrawing them from

the sludge hopper and pumping them

back to the center well of the thickener,
© (Continued on page 218)
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